Dear Editor:

Thank you for reviewing my manuscript, "Integrating Disulfdidptosis-Related Long Noncoding RNAs in Colorectal Cancer Prognosis: A Path to Precision Medicine." We appreciate the thorough evaluation conducted by the Editorial Office and the peer reviewer.

I am pleased that the preliminary decision indicates my manuscript meets the publication standards of the World Journal of Clinical Oncology. I value the constructive feedback from Reviewer #1 and am committed to addressing his suggestions to enhance the scientific quality and clarity of our work.

1. Specific statements for future perspectives should be added.

Response: I have included specific statements for future perspectives, highlighted in green as the last paragraph.

2. The current summary comment is too abstract and this comment does not make Disulfdidptosis-related long non-coding RNAs in colorectal cancer prognosis look good.

Response: I have revised the summary marked with green colour in the revised manuscript to clarify and emphasize the importance of Disulfdidptosis-related long non-coding RNAs in colorectal cancer prognosis. Specifically, I highlighted the implications of drug sensitivity analysis, stressing the need for personalized treatment strategies to enhance efficacy.

Thank you for this opportunity, and please don't hesitate to contact me for any further clarification.

Sincerely,

Corresponding author: Shi-Yan Zhang (myebox@139.com)

2024-03-05